Author's response to reviews

Title: Bronchial obstruction secondary to idiopathic scoliosis in a child: a case report

Authors:

Saad Alotaibi (saalotai@hotmail.com)
James Harder (james.harder@calgaryhealthregion.ca)
Sheldon Spier (Sheldon.Spier@CalgaryHealthRegion.ca)

Version: 11 Date: 20 May 2008

Author's response to reviews: see over
The manuscript was edited according to the editors advice:
1. The consent was moved as advised after the authors contributions section.
2. There was no acknowledgment for this manuscript so this was not needed.
3. There were no funding for this manuscript so this was not added.
4. The table was moved from the figures section and added to the body of the manuscript and it was edited according to the reviewer request using simple 1 style.
5. The figures are in separate pages and with no comments or numbering according to your advice.
6. The manuscript was reviewed for final submission and there were no changes.
This is our final submission and we believe it is ready for publication after your permission.
Best regard.